Author's response to reviews

Title: Antihyperlipidemic Effects of Pleurotus Ostreatus (Oyster Mushrooms) in HIV-Infected Individuals Taking Antiretroviral Therapy

Authors:

Donald I Abrams (dabrams@hemeonc.ucsf.edu)
Paul Couey (pcouey@hemeonc.ucsf.edu)
Starley B Shade (starley.shade@ucsf.edu)
Mary Ellen Kelly (mkelly@hemeonc.ucsf.edu)
Nnemdi Kamanu-Elias (nnemdi@gmail.com)
Paul Stamets (paulstamets@gmail.com)

Version: 2 Date: 9 January 2011

Author's response to reviews:

January 9, 2010
Jigisha Patel, MRCP, Ph.D.
Editor
BMC Complementary and Alternative Medicine

Dear Dr. Patel:

RE: Antihyperlipidemic Effects of Pleurotus Ostreatus (Oyster Mushrooms) in HIV-Infected Individuals Taking Antiretroviral Agents

We are pleased to resubmit our manuscript after making the recommended changes suggested by Iratxe Puebla. We have reformatted the abstract to conform to Background, Methods, Results and Conclusions. We had already ended the abstract section with the Trial Registration section and number. The IRB information was added to the methods section of the document. Competing Interests and Authors' Contributions sections were added at the end of the manuscript.

Thank you for the opportunity to make this corrections and modifications of our submission.

We appreciate your consideration of the manuscript and hope we can look forward to additional constructive peer review.

Wishing you the best for the New Year,

Donald I. Abrams, MD
Professor of Clinical Medicine
University of California San Francisco